Author's response to reviews

Title: A systematic review and meta-analysis of neurological soft signs in relatives of people with schizophrenia

Authors:

Kishen Neelam Dr (kishen.neelam@postgrad.manchester.ac.uk)
Deepak Garg Dr (deegarg@gmail.com)
Max Marshall Professor (max.marshall@manchester.ac.uk)

Version: 2 Date: 30 April 2011

Author's response to reviews:

Please see attached cover letter.